Author's response to reviews

Title: Immuno-epidemiology of human Schistosoma haematobium infection: preferential IgG3 antibody responsiveness to a recombinant antigen dependent on age and parasite burden.

Authors:

Francisca Mutapi (f.mutapi@ed.ac.uk)
Takafira Mduluza (tmduluza@yahoo.com)
Natalia Gomez-Escobar (ngescobar@mrc.gm)
William B Gregory (b.gregory@ed.ac.uk)
Cecilia Fernandez (cfern@fq.edu.uy)
Nicholas Midzi (nmidzi@nihr.co.zw)
Rick M Maizels (r.maizels@ed.ac.uk)

Version: 3 Date: 8 June 2006

Author's response to reviews:

The following changes have been made to the manuscript:

1) William B Gregory changed to William F Gregory.
2) Deleted the word 'Footnote' from page 1.
3) Abstract has been structured into Background, Methods, Results, and Conclusions.
4) Materials & Methods - renamed to 'Methods'.
5) References 10 and 16 now contain all authors' names
6) Figure title removed from the image files.